Author's response to reviews

Title: Evaluation of a Reproductive Health Awareness Program for Adolescence in Urban Tanzania - A Quasi-experimental Pre-test Post-test Research -

Authors:

Frida Madeni (skuiyo@yahoo.com)
Shigeko Horiuchi (shigeko-horiuchi@slcn.ac.jp)
Mariko Iida (09DN001g7G-ts@slcn.ac.jp)

Version: 5 Date: 25 June 2011

Author's response to reviews:

We deleted the citation of Table 6 in the method section by changing the expression. Now the tables are numbered consecutively according to their order of appearance in the text.